Author’s response to reviews

Title: SIRT1 (rs3740051) role in pituitary adenoma development

Authors:
Rasa Liutkeviciene (rasa.liutkeviciene@lsmuni.lt)
Alvita Vilkeviciute (alvita.vilkeviciute@lsmuni.lt)
Greta Morkunaite (greta.morkunaite@lsmuni.lt)
Brigita Glebauskiene (brigita.glebauskiene@lsmuni.lt)
Loresa Kriauciuniene (loresa.kriauciuniene@lsmuni.lt)

Version: 6 Date: 05 Sep 2019

Author’s response to reviews:

Dear Editor,

Thank you for the revision. We made the corrections according to your comments

1. Tables

We added the information from Table 1 and Table 2 in our manuscript.

2. Competing interests

Competing interests statement was corrected.

3. Abbreviations

All abbreviations have been included in the ‘Abbreviations’ section.

4. Attachment entitled ‘Reply to reviewers comments 2019-09-03.docx’ was deleted.

Best regards, Alvita Vilkeviciute